Author's response to reviews

Title: Study Protocol for a Comparative Effectiveness Trial of Two Parent Training Programs in a Fee-for-Service Mental Health Clinic: Can We Improve Mental Health Services to Low-Income Families?

Authors:

Deborah A Gross (debgross@jhu.edu)
Harolyn ME Belcher (belcher@kennedykrieger.org)
Mirian E Ofonenedu (ofonenedu@kennedykrieger.org)
Susan M Breitenstein (susan_m_breitenstein@rush.edu)
Kevin Frick (kfrick@jhu.edu)
Budhathoki Chakra (cbudhat1@jhu.edu)

Version: 4 Date: 31 January 2014

Author's response to reviews: see over
Author’s response to reviews

Title: Study Protocol for a Comparative Effectiveness Trial of Two Parent Training Programs in a Fee-for-Service Mental Health Clinic: Can We Improve Mental Health Services to Low-Income Families?

Authors:

Deborah A Gross (debgross@jhu.edu)
Harolyn ME Belcher (belcher@kennedykrieger.org)
Mirian E Ofonedu (ofonedu@kennedykrieger.org)
Susan M Breitenstein (susan_m_breitenstein@rush.edu)
Kevin Frick (kfrick@jhu.edu)

Version: 4 Date: 30 January 2014

Author’s response to reviews:

We noted some errors in the description of the analyses and have corrected those in the attached revision. We have also added an author.